Author's response to reviews

Title: Spanish version of the Oral Health Impact Profile (OHIP-Sp)

Authors:
   Rodrigo Lopez (rlopez@odont.au.dk)
   Vibeke Baelum (baelum@odont.au.dk)

Version: 4  Date: 7 July 2006

Author's response to reviews:

The BioMed Central Editorial Production Team

July 7, 2006

1. In the submission system, Abstract section, we have renamed "Results and Discussion" section to "Results".

2. We have referred to "Additional files" as indicated.

3. We have now removed underlining of subtitles in pages 6, 9 and 13.

Sincerely yours

Rodrigo Lopez